Author’s response to reviews

Title: Implementation of genotype-guided dosing of warfarin with point-of-care genetic testing in three UK clinics: a matched-cohort study

Authors:
Andrea Jorgensen (a.l.jorgensen@liverpool.ac.uk)
Clare Prince (cprin@liverpool.ac.uk)
Gail Fitzgerald (Gail.FitzGerald@liverpool.ac.uk)
Anita Hanson (africa98@liverpool.ac.uk)
Jenny Downing (J.Downing@liverpool.ac.uk)
Julia Reynolds (Julia.Reynolds@innovationagency.nwc.nhs.uk)
J. Zhang (Eunice.Zhang@liverpool.ac.uk)
Ana Alfirevic (zagreb@liverpool.ac.uk)
Munir Pirmohamed (munirp@liverpool.ac.uk)

Version: 2 Date: 15 Mar 2019

Author’s response to reviews:

Professor Andrea Jorgensen
Department of Biostatistics
Institute of Translational Medicine
University of Liverpool
Liverpool
L69 3GL

14/3/2019

Dear Editors
Sincere thanks for your response. We have addressed the editorial comments and provided a response to each below. We hope that you will be able to consider our manuscript further for publication in your journal, and look forward to hearing from you soon.

Editorial comments

1. Additional files should be uploaded as a “Supplementary” file type in the system. At the moment, supplemental files are present in the main manuscript and they should be uploaded as supplementary files

   - We apologise for this oversight, and have now uploaded the supplementary files as additional files, removing them from the main manuscript.

2. The abstract should be formatted as Background/Methods/Results/Conclusions, and be no more than 350 words. Please change interpretation section to conclusions

   - We have now changes ‘interpretation’ section to ‘conclusions’ section in the Abstract.

3. Please add three to ten keywords under the abstract

   - Keywords have now been included under the Abstract.

4. Please move all URLs from the main text to the reference list, and cite them accordingly. Please remove URL from page 7 and reference accordingly

   - We apologise for this oversight, and have removed the URL from page 6, line 152, referencing accordingly.
Thank you again for considering our manuscript.

Sincerely,

Professor Andrea Jorgensen